Author's response to reviews

Title: Knowledge of breast cancer and its'early detection measures among rural women in Akinyele Local government Area in Ibadan, Nigeria

Authors: O Abimbola Oluwatosin (aoluwatosin@yahoo.com)
Oladimeji Oladepo (oladepod@yahoo.com)

Version: 6 Date: 23 November 2006

Author's response to reviews:

Authors: O Abimbola Oluwatosin and Oladimeji Oladepo

Title: Knowledge of breast cancer and its'early detection measures among rural women in Akinyele Local government Area in Ibadan, Nigeria

Journal: BMC Cancer

MS: 1092922888110792

The formatting changes requested have been done. All the additional files have been removed as they were not refered to in the text. Some changes were made, Abstract: under methods, 407 was changed to 420. Reference numbers were changed within text to allign with reference list. Thank you for publishing the paper.

Dr Abimbola Oluwatosin